Author's response to reviews

Title: Improved quality of management of eclampsia patients through criteria based audit at Muhimbili National Hospital, Dar es Salaam, Tanzania. Bridging the quality gap

Authors:

Hussein L Kidanto (hkidanto@yahoo.co.uk)
Peter T Wangwe (drpeterwangwe@yahoo.com)
Charles D Kilewo (kilewo1950@yahoo.com)
Lennarth Nystrome (Lennarth.Nystrom@epiph.umu.se)
Gunilla Lindmark (Gunilla.Lindmark@kbh.uu.se)

Version: 6 Date: 8 November 2012

Author's response to reviews:

PO box 65439
Dar es Salaam Tanzania
08th November, 2012

The Editor,
BMC-Pregnancy and Childbirth

Dear Sir/Madam,

RE: COVER LETTER

We wish to re-submit our manuscript titled: Improving quality of management of eclampsia patients through criteria based audit at Muhimbili National Hospital, Dar es Salaam, Tanzania. Bridging the quality gap. to be considered for publication in your widely read open access Journal.

We have made changes as suggested by the reviewer, with this covering letter we attach :-

1. Response to reviewers: We have responded point by point to all reviewers comments.

2. Revised manuscript: This has been revised as suggested by reviewers

With this covering letter we thank you in advance

Best Regards,
Kidanto HL. (Corresponding author)